
\documentstyle[12pt]{article}
\textwidth 15true cm
\textheight 8.65true in
\oddsidemargin 0.25true in
\evensidemargin 0.25true in
\topmargin -0.3true in
\headsep 0.4true in

\begin{document}

\begin{flushright}
SCI199Y: September 19, 1995
\end{flushright}

\bigskip

\noindent{\bf Additional reading on graphical techniques}
\begin{itemize}
\item {\it Envisioning Data.} E.R. Tufte (1991) Cheshire: Graphics Press. 
\item {\it The Elements of Graphing Data.} W.S. Cleveland (1994)
New Jersey: Hobart Press. [2nd edition; 1st edition 1985]
\item {\it Visualizing Data.} W.S. Cleveland. (1994) New Jersey: Hobart Press.
\item {\it Visual Realizations.} H. Wainer.  A regular column in the monthly
magazine {\it Chance}.
\end{itemize}

\noindent{\bf In the news this week...}
\begin{itemize}
\item ``Cancer deaths soar among women smokers": Globe and Mail, Sept. 16.
Two new studies show that lung-cancer death rates among North American women 
smokers are out of control.  In Canada, the rate this year is expected
to outstrip that of breast cancer, while in the U.S., fatalities were found
to have increased sixfold between the sixties and the eighties.
\item ``Even mild weight gain threatens life, study finds": Globe and Mail,
Sept. 14; [with accompanying graphic] also reported on the National, Sept. 13.
\item ``Consumer price index falls 0.1\%'': Globe and Mail, Sept. 16.
[with accompanying graphic]
\end{itemize}

\noindent{\bf Required for next week}
\begin{itemize}
\item Anything that didn't get done for this week.
\item Read Chapters 2 and 3 in {\it VDQI}.  
Come to class prepared to note the parts you didn't follow, 
and the parts you found particularly interesting.
\end{itemize}

\end{document}


 
